Author’s response to reviews

Title: Preventive Evidence into Practice: What factors matter in a facilitation intervention to prevent vascular disease in family practice?

Authors:

Grant Russell (grant.russell@monash.edu)

Riki Lane (riki.lane@monash.edu)

Sharon Parker (sharon.parker@unsw.edu.au)

John Litt (jlitt@flinders.edu.au)

Danielle Mazza (danielle.mazza@monash.edu)

Jane Lloyd (j.lloyd@unsw.edu.au)

Nicholas Zwar (nzwar@uow.edu.au)

Mieke van Driel (m.vandriel@uq.edu.au)

Chris Del Mar (cdelmar@bond.edu.au)

Jane Smith (jsmith@bond.edu.au)

Mark Harris (m.f.harris@unsw.edu.au)

Version: 1 Date: 17 Nov 2018

Author’s response to reviews:

Dr Tovah Honor Aronin, Ph.D.
Editor
BMC Family Practice
Dear Dr Aronin
Re Response to reviewers comments: submission to FAMP-D-18-00265

Thank you for your advice and the reviewers’ comments provided to our paper Preventive Evidence into Practice: What factors matter in a facilitation intervention to prevent vascular disease in family practice? (Ref FAMP-D-18-00265)
We have made a substantial revision of the manuscript using the insights and suggestions provided by your reviewers. A number of the suggestions have asked for more information, and, in some cases for additional references. In our response, we have created a response table where comments have been grouped by section of the paper, indicating where each individual reviewer has requested amendment. We hope that this approach makes it easier to follow the changes that we have made. Note that these responses and a letter to the editor have been included as supplementary files - I hope that this is satisfactory to your process. It can be amended if you require

Yours sincerely

Grant Russell

Professor of Primary Care Research